Check for updates

OPEN ACCESS

EDITED BY Kensuke Harada, Osaka University, Japan

REVIEWED BY

Gustavo Alfonso Garcia Ricardez, Ritsumeikan University, Japan Mehdi Benallegue, UMI3218 Joint Robotics Laboratory (JRL), Japan

*CORRESPONDENCE Shakra Mehak, Smehak@pilz.ie Inês F. Ramos, ines.fernandes@unimi.it Maria Chiara Leva, mariachiara.leva@tudublin.ie

RECEIVED 17 May 2024 ACCEPTED 06 November 2024 PUBLISHED 12 December 2024

CITATION

Mehak S, Ramos IF, Sagar K, Ramasubramanian A, Kelleher JD, Guilfoyle M, Gianini G, Damiani E and Leva MC (2024) A roadmap for improving data quality through standards for collaborative intelligence in human-robot applications. *Front. Robot. Al* 11:1434351. doi: 10.3389/frobt.2024.1434351

COPYRIGHT

© 2024 Mehak, Ramos, Sagar, Ramasubramanian, Kelleher, Guilfoyle, Gianini, Damiani and Leva. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

A roadmap for improving data quality through standards for collaborative intelligence in human-robot applications

Shakra Mehak^{1,2}*, Inês F. Ramos³*, Keerthi Sagar⁴, Aswin Ramasubramanian⁴, John D. Kelleher⁵, Michael Guilfoyle¹, Gabriele Gianini⁶, Ernesto Damiani³ and Maria Chiara Leva²*

¹Pilz Ireland Industrial Automation, Cork, Ireland, ²School of Food Science and Environmental Health, Technological University Dublin, Dublin, Ireland, ³Secure Service-oriented Architectures Research Lab, Department of Computer Science, Università degli Studi di Milano, Milan, Italy, ⁴Robotics and Automation Group, Irish Manufacturing Research Centre, Mullingar, Ireland, ⁵School of Computer Science and Statistics, Trinity College, Dublin, Ireland, ⁶Department of Informatics, Systems and Communication (DISCo) Università degli Studi di Milano-Bicocca, Milano, Italy

Collaborative intelligence (CI) involves human-machine interactions and is deemed safety-critical because their reliable interactions are crucial in preventing severe injuries and environmental damage. As these applications become increasingly data-driven, the reliability of CI applications depends on the quality of data, shaping the system's ability to interpret and respond in diverse and often unpredictable environments. In this regard, it is important to adhere to data quality standards and guidelines, thus facilitating the advancement of these collaborative systems in industry. This study presents the challenges of data quality in CI applications within industrial environments, with two use cases that focus on the collection of data in Human-Robot Interaction (HRI). The first use case involves a framework for quantifying human and robot performance within the context of naturalistic robot learning, wherein humans teach robots using intuitive programming methods within the domain of HRI. The second use case presents real-time user state monitoring for adaptive multi-modal teleoperation, that allows for a dynamic adaptation of the system's interface, interaction modality and automation level based on user needs. The article proposes a hybrid standardization derived from established data quality-related ISO standards and addresses the unique challenges associated with multi-modal HRI data acquisition. The use cases presented in this study were carried out as part of an EU-funded project, Collaborative Intelligence for Safety-Critical Systems (CISC).

KEYWORDS

human robot interaction (HRI), collaborative intelligence, ISO standard, human machine interaction, artificial intelligence, machine learning, ISO 8000

1 Introduction

The emergence of CI with Industry 4.0 represents a significant convergence between human and Artificial Intelligence (AI) abilities, becoming one of the main focus of the transition to Industry 5.0 Demir et al. (2019). Particularly in the field of collaborative robotics, where humans, robots and AI work in collaboration alongside each other, leading to a new age of enhanced productivity, adaptability and innovation Villani et al. (2018). The term CI refers to the execution of shared tasks by humans and AI systems in which they exploit the strengths of each other to increase productivity Toniolo et al. (2023). This concept is often studied under different terminologies, human-centric AI Rožanec et al. (2023), human-machine-in-loop Mosqueira-Rey et al. (2023) and human-AI interaction Amershi et al. (2019). However, the current immaturity of certain capabilities of AI, such as lack of intuitive reasoning, creativity, common sense, and situational analysis, as well as limited capabilities of humans in tasks that require long-term memory, repetition, speed, and accuracy, restrict the application of either full automation or solely manual operations to complex production environments (Villani et al., 2018). Furthermore, collaboration enables flexible and adaptive access to system resources, resulting in a more advantageous solution than swapping between human and machine tasks (Angleraud et al., 2021; Zhou et al., 2022).

One of the fundamental applications of the CI system is Human-Robot Interaction (HRI), which allows humans and robots to work together as a team (Panagou et al., 2024) and complement each other's skills. The implementation of HRI systems necessitates the integration of heterogeneous data, including various data sources from humans and robots. This includes data on robot dynamics and kinematics, system and process state data from internal repositories, workspace and environment data from extrinsic sensors, humanrelated data from wearable and other sensors, human feedback from controllers or interfaces, and intelligent predictive or prescriptive information from an AI component. Moreover, these systems rely on real-time sensor data from the environment and human activity to recognize changes in the environment, enabling the customization of robot behaviour in accordance with operator preferences (Heinisch et al., 2024; Gao et al., 2021; Gast et al., 2009). To build a smooth interaction between humans and robots, a high quality multimodal data is required to be processed to design and develop a synergistic HRI system. There is comprehensive evidence demonstrating the correlation between data quality and AI model performance in the literature. However, data quality's role in CI system development and performance is a less explored area but equally critical in collaborative systems because of the interactive nature of CI (Amershi et al., 2019; Rožanec et al., 2023; Gudivada et al., 2017). The main attributes of CI systems are accurate and timely decision-making (Mentzas et al., 2021), transparency (Vössing et al., 2022), user trust (McGrath et al., 2024) and poor data quality may result in mismatched in the human-machine interactions and inaccurate and untrustworthy system behaviours. Empirical studies on CI systems show that data quality directly affects system performance (Li et al., 2023), however, the extent of data quality's consequence on CI systems remains an open ground for further research.

Data quality has been thoroughly explored beyond the HRI fields and has gained significant attention from the industry due to the multifaceted value that data can bring to the business (Batini et al., 2009). This effort can be observed by the development of data and data quality-related standards, namely ISO/IEC 25000 series (SQuaRE) (ISO/IEC 25000, 2014a), which describes requirements and evaluations for the quality of systems and software, and (ISO 8000-61, 2022a), which details data management processes. Despite this, the lack of standards practice and inadequacy in data-related standards (Odong et al., 2022) is a substantial constraint in achieving reliable CI systems. In this regard, this study aims to contribute to the literature by providing a simple framework for the standardization of data processes and data quality in HRI.

This article contributes as follows.

- The paper describes the key data quality challenges specific to multimodal HRI data acquisition for CI application, dissecting the complexities of data collection from various sensors and input types in industrial scenarios, also considering data obtained from interactions between humans and robots as well as, metrics related to physiological measures indicative of human cognitive and physical state, and subjective responses.
- Review of existing ISO standards and guidelines specifically addressing data quality for multimodal HRI data acquisition and the gaps identified for their applicability and their capacity to cover all data quality issues in multimodal HRI data.
- Hybrid standardization approach and a five-step data quality management plan based on ISO standards that can be effectively adapted for CI applications.
- Demonstration of the application and efficacy of the proposed hybrid standardization approach in managing data quality within two specific case studies in industrial HRI.

The paper is structured as follows: Section 2, provides a comprehensive background, which introduces collaborative intelligence, human-robot interaction, and an overview of data quality standards relevant for it. Section 3 presents a hybrid standardization for CI. Section 4 discusses two HRI use cases, and presents them to highlight the data collection methodology applied to the case studies and the issues related to data quality compared to the "Data Management Steps for CI applications" proposed in section 3. Section 5 summarizes the contributions, discusses their implications, and proposes potential avenues for future research.

2 Background and literature review

2.1 Collaborative intelligence (CI)

As previously discussed, CI is the seamless integration of human cognitive abilities with AI systems in order to facilitate decision-making, problem solving, and development in many different areas. Unlike conventional automation paradigms that aim to replace human work with machines, CI promotes collaboration between humans and AI systems. This approach takes advantage of the unique strengths of both entities to produce results that neither could independently achieve (Dellermann et al., 2021; Schleiger et al., 2024).

The implications of CI are disruptive in industries, as it improves efficiency and safety. For example, CI systems merge human insights with AI to optimize production demands, streamline the process, and reduce production time (Trakadas et al., 2020) in the manufacturing sector. Furthermore, it improves workplace safety by integrating human supervision with AIpowered surveillance technologies to monitor and predict safety incidents (Zheng et al., 2023).

The evolution of CI has been accompanied by important technological developments, particularly in the domains of machine learning (Moustakis and Herrmann, 1997; Semeraro et al., 2023), data analytics (Nardo et al., 2020), and Internet of Things (Bissoli et al., 2019). Current advances in the field of CI emphasize the necessity of adaptive learning (Sharifi et al., 2021) that can improve system responsiveness based on user feedback (Mehta and Losey, 2023), supporting sophisticated and contextually sensitive collaboration between human and AI systems. Moreover, CI in an industrial setting faces a plethora of challenges, starting with the need for heterogeneous data from multiple sources (Victorelli et al., 2020) subjective and contextual user feedback, the privacy and confidentiality of data streams (Karat et al., 2007) and data quality management. Addressing these challenges is critical if one wants to exploit CI systems to their full potential. This study focuses primarily on the quality of the data, which serves as the foundation for successful human-AI collaboration.

2.2 Data quality

The role of data quality cannot be overstated in many fields. Several works have been published on data quality challenges and management in AI models, machine learning (ML), and Internet of Things systems. Studies by (Ehrlinger and Wöß, 2022; Gong et al., 2023; Zhou et al., 2021) highlight the challenges in achieving high-quality data in ML, pointing out that the complexity of data sources often outpace current standardization efforts, such as those proposed by (ISO/IEC 25012, 2008a). Recently, (Eckman et al., 2024) has introduced the application of survey methodologies to improve data collection processes, suggesting that the strategies developed for survey data quality can be adapted to improve ML data acquisition. By examining data quality aspects within ML pipelines, including data collection, pre-processing, and validation, (Rangineni, 2023) emphasize the imperative of adhering to quality standards in ML development pipelines, exploring the direct correlation between data quality and model performance. Furthermore, (Gudivada et al., 2017) proposed a data quality management framework for big data and ML, highlighting the importance of standards to ensure data integrity in a large-scale data environment. The authors in (Gupta et al., 2021) discuss the lack of focused efforts to improve data quality in current research practices and underscore the fundamental limitation that the performance of ML models is bounded by the quality of the data used for training. Another work by (Perez-Castillo et al., 2018) proposed a (ISO 8000-61, 2022a) based quality management framework for sensor data to overcome the challenges associated with data quality activities in smart systems. Moreover, the authors in (An et al., 2020) investigate the application of blockchain technologies as a method to improve data integrity and security in IoT systems. Their work presents a decentralized approach to data quality management, using blockchain's inherent properties to ensure reliability of IoT data.

Collectively, these studies highlight the importance of data and dataset quality in system development and functionality, and advocate for a multidisciplinary approach, combining insights from data science, ML and quality management to develop strategies to ensure data reliability in complex computational environments.

2.3 Human robot interaction and data quality challenges

The field of HRI is significant within the broader domain of collaborative intelligence. It is dedicated to understanding, developing, and refining the interfaces and interactions between humans and robotic systems, ensuring that such collaborations are as effective and efficient as possible. The type of interaction includes both direct physical interaction and shared decisionmaking processes, in which both humans and robots contribute their respective abilities. Moreover, HRI exceeds traditional automated systems, introducing robots capable of working alongside human operators in shared spaces (Hentout et al., 2019).

The development of human data-driven HRI systems has resulted in notable advances in productivity and safety within the sector (Liu et al., 2016; Saheb Jam et al., 2021; Admoni and Scassellati, 2014; Ravishankar et al., 2024; Yang et al., 2021). In addition, the utilization of human feedback insights such as cognitive or emotional state data (Khamassi et al., 2018), visual data (Gao et al., 2021), and direct communication cues (Khamassi et al., 2018) enable HRI systems to adapt and respond to the needs and safety of human operators. In addition, these data can be used in the training of adaptive models to allow robots to adjust their behaviors and functionality to improve collaboration (Skantze et al., 2014). Through understanding of the stress levels, concentration, and fatigue experienced by human operators, robots possess the ability to independently adapt to or notify humans of potential hazards, creating a working environment that is reliable and responsive (Bekele and Sarkar, 2014). Furthermore, the successful implementation of HRI is based on the precision, dependability, and timeliness of the data used in the development and training of the model.

In HRI systems the quality of the data depends on the purpose of their use, however, common data quality challenges include: (a) the operationalization of relevant human signals or feedback (Kim et al., 2020), (b) the distinction between noise, confounding effects, and individual variation, (c) the prevention or correction of missing data due to sensor malfunction (Teh et al., 2020), recording or storing issues, loss of synchronization or human agent non-compliance with the data collection process (Celiktutan et al., 2017), or (d) the capture of relevant environmental, workplace and task contexts that impact the operational design domain and ecological validity of the system functions (Hoffman and Zhao, 2020; Han and Williams, 2022). In this context, the impact of data quality degradation can be measured by its impact on HRI quality, as assessed by different robot-related metrics, human-related metrics or human-robot interaction-related metrics (George Kokotinis et al., 2023).

2.4 An overview of ISO standards regarding data quality

ISO standards explicitly referring to data quality are significant within industrial systems, as they establish a structure for evaluating, controlling, and improving data quality throughout its entire lifetime. These standards set forth criteria and guidelines to ensure that data, regardless of their origin, meet the strict criteria necessary for precise analysis and decision-making. In the context of CI systems, where decisions are based on data-driven AI analysis and human input, adherence to these standards is imperative for CI data. Table 1 summarizes key international data standards that present data quality frameworks and management protocols that can be suitable for CI data. To demonstrate the importance of data quality standards for collaborative data, we have also specified their impact level based on their contribution to the foundational quality (Perez-Castillo et al., 2018; Batini et al., 2009; Panduwiyasa et al., 2021) and expert opinion. For instance, ISO 8000-61 (ISO 8000-61, 2022a) is marked as high due to its critical role in establishing the data quality framework, and it is widely adopted and highly endorsed for its comprehensive approach to data quality (Perez-Castillo et al., 2018).

Among these, the ISO 8000 series plays a crucial role in the context of data quality management by providing a comprehensive set of standards. These standards are designed to effectively monitor data quality in various business operations. Specifically (ISO 8000-1, 2022b), emphasizes the importance of data quality in achieving specific goals and objectives. The standard promotes a structured approach to data quality and governance to improve the decisionmaking process. According to this standard, data quality is described as an objective concept that can be assessed based on how well the data aligns with its intended purpose. By offering a wideranging framework, (ISO 8000-1, 2022b) addresses data quality across different data types, formats, and organizational settings. It can also be applied to leverage data in operational activities and model training for ML systems, which can be relevant for CI systems. However, the standard needs more detailed information on the specific processing methods involved. Furthermore (ISO 8000-8, 2015a), defines and measures the quality of information and data quality based on three categories: syntactic, semantic and pragmatic quality. Syntactic quality measures conformance to syntax requirements, semantic quality measures data alignment to what it represents in the system, and *pragmatic* quality measures the suitability of the data for a particular application. For CI data, semantic quality verification can involve, (a) Identifying any missing data due to sensor malfunction, (b) data acquisition issues, (c) loss of synchronization or human agent non-compliance with the data collection process, and (d) ensuring that data variances match expected patterns to appropriately map data to entities in the operational domain (ISO 8000-8, 2015a). The practical relevance of data in CI contexts depends on its pragmatic quality, which considers the usefulness of the data against costs and benefits, particularly important when using hard-to-collect human data. The ISO 8000-61 (ISO 8000-61, 2022a) presents a structured data quality management approach. The reference model begins with comprehensive data quality planning, requirement specifications, rigorous data processing, monitoring, and management. The framework for considering the characteristics of the CI data in the present study is specifically derived from the ISO 8000-61 (ISO 8000-61, 2022a).

The ISO 25000 (ISO/IEC 25000, 2014a) series, addresses the Systems and Software Quality Requirements and Evaluation (SQuaRE). They offer a comprehensive framework to guarantee the quality of software products and systems, mainly those relevant to ML and potentially CI systems. The key standards of this series include (ISO/IEC 25001, 2014b), which handles quality management; ISO 25012 (ISO/IEC 25012, 2008a), which focuses on the data quality model; and (ISO/IEC 25030, 2019), which deals with quality requirements. ISO 25012 (ISO/IEC 25012, 2008a) defines a data quality model required to assess data quality needs across multiple data processes such as production, collection, and integration, emphasizing the importance of data quality within compliance guidelines. It covers different data quality attributes needed to maintain the integrity and practicality of data within machines and human interaction. These factors, such as correctness, completeness, consistency, credibility, and efficiency, are essential to ensure the reliability and effectiveness of the data (ISO 8000-61, 2022a). As human data is commonly used within the scope of CI, the concepts of compliance, efficiency, traceability, and accessibility are significant for human factor impact assessment Schleiger et al. (2024). In addition (ISO/IEC 25010, 2023a), introduces a quality model for software that extends beyond data to incorporate the quality of use, focusing on quality assurance and used in conjunction with (ISO/IEC 25012, 2008b). To achieve optimal results, it is critical to take into account both the data and the application environment, as this perspective is fundamental to CI applications.

ISO/IEC 22989 (2023b) establishes a framework for AI through the definition of key concepts and terminology that encompass various aspects of AI, such as systems and data. It underscores the importance of data precision, the process of training and validating ML models and the iterative nature of retraining models, and highlights the ongoing evolution of AI systems to adapt and improve themselves over time. Furthermore, the standard emphasizes the critical role of data quality, annotation, augmentation, and the significance of data sets in the training, validation, and testing of AI models, thus influencing the efficacy and reliability of AI systems. The lifecycle of an AI system comprises multiple stages, with the standard stressing the pivotal role of data at each phase, for example, training, validation, and testing. These stages are fundamental in the evolution of AI models, necessitating the use of high-quality, varied, and inclusive datasets to construct models that function dependably and impartially in real-world settings. By offering a vocabulary for discussing the AI system's lifecycle, the standard facilitates the integration of AI into collaborative environments, ultimately supporting the adoption of CI systems. Similarly (ISO/IEC 23053, 2022c), offers a systematic framework for data management in ML systems. This standard focuses on the key stages of data collection, preprocessing, and validation to ensure the efficiency and dependability of AI models. The procedure highlights the importance of precise data classification by segregating training, validation, testing, and production data sets while ensuring statistical uniformity to enhance model evaluation and practicality.

TABLE 1 Summary of the key data quality standards for HRI systems.

Standard	Title	General description	Relevance to data quality	Relevance to CI/HRI context	Impact
IEC 8000-61 ISO (2022a)	Data Quality Management: Process reference modal	Provides a process reference model framework for data quality management, including principles for data quality planning and control	Fundamental to measure the quality of collected data and improves the processes to avoid data nonconformity	Suitable for managing heterogeneous data sources and identifying potential risks associated with data or system	High
ISO/IEC 25010 ISO (2023a)	System and Software Engineering	Defines software product quality model requirement elicitation and specification	Integral to quality testing and quality assurance	This can be applied to define CI data quality in use, regarding the effectiveness, productivity, safety, and satisfaction of the user interaction with the system	Medium
ISO/IEC 25022 ISO (2008b)	System and Software Engineering	Defines quality in use, intended to be use with other software quality related standards	Provides evaluation process and normalizing quality in use assessment	Offers similar role like ISO (2023a)	Low
Standard	Title	General Description	Relevance to Data Quality	Applicability to CI/HRI Context	Significance
ISO/IEC 25012 ISO (2008a)	Software Engineering	Defines a quality model for data used in software systems, covering 15 data quality characteristics	Address data quality characteristics	The quality characteristics can be used for data quality assessment of captured user feedback, interaction experience and data interpretations	High
ISO/IEC 25024 ISO (2015b)	System and Software Engineering	The standard provides set of specific quality measures for each related components and corresponding entity	Defines quantitative measures to assess data quality and describes an explanation of applying those measures for quality assessment	The quantitative measures can be used to analyze patterns and correlations of collected multimodal data for the implementation and development of the HRI system	Medium
ISO/IEC 25030 ISO (2019)	Software Engineering Quality Requirements	Focuses on the requirements for software product or data quality, use for requirement definition, use and govern	The offered framework in this standard use to evaluate that the data meets the defined and expected quality criteria	Supports in determining clear contexts in which the CI system will operate and ensuring optimal functionality in different conditions	Low
ISO/IEC 23053 ISO (2022c)	Framework for AI System Using ML	Offers data acquisition and data preparation steps in the ML pipeline	Pertinent for the development and maintenance of AI components in CI systems	The standard supports the development, deployment, and management of AI systems within collaborative environments, ensuring that these systems are built and operated transparently and effectively	High

The standard emphasizes the processes of cleaning and normalizing the input-output data arrangements to enhance the performance of the model. It is important to note that complying with the systematic methodology for data management outlined in (ISO/IEC 23053, 2022c) is a prerequisite for developing AI systems that demonstrate the ability, dependability, and adaptability required for certain application domains.

3 Hybrid standardization

Hybrid standardization refers to a systematic approach to identifying related standards that combines the strengths of system-driven selection and collaborative preference among stakeholders (Van Wegberg, 2004). It allows the re-contextualization of several standards to better cover the emerging needs of complex systems with multiple interacting components.

This study proposes a hybrid standardization framework to improve data quality, specifically in CI systems. A CI system, as depicted in Figure 1, is composed of several subsystems, including the Computer/AI/ML subsystem(s) that stores, manages, and employs both "target ML data" and "non target data," distinguishing the type of data. The information and interface subsystem is responsible for user communication and between-systems communication functions which is an essential component of CI systems. Other subsystems include human collaborators, robotic or mechanical subsystems, sensor subsystems, and environmental components. Data streams from any of these sub-components, often from more than one, can be used by the ML/AI component or as contextual supportive information to implement the required CI functions.

The framework takes into account the changing nature of CI data, which involves complex interactions with human-generated content that is subjective and context-dependent. Moreover, it uses current data standards with best practices in ML data management and identifies and applies specific standards to two main data categories within the CI system. They are "target ML data" and "non-target data." For "target ML data," the framework implements extensive standards application to maintain accuracy, completeness, and consistency. For "non-target data," it allows for flexibility to handle variability and contextual differences, maintaining a quality baseline to support CI functionalities. The implementation involves systematic mapping of ML or machine data flows to adhere to data quality protocols, from collection through processing to the formation of a well-curated dataset. Following that, it offers a process guide of five steps for applying the proposed framework to achieve data quality in CI systems, presented in Figure 2 based on the guidelines from the standards. This hybrid approach provides general management strategies to promote quality throughout the entire lifecycle of the data, which can be extended to CI dataset creation and use in complex industrial environments. The process is divided into two main phases. The first phase can be applied before data collection or acquisition and theoretically before system/model development. This phase determines the specification of quality requirements and quality planning, establishing a plan for data quality risk mitigation and management and followed by the second phase of data quality preparation and control. The second phase aims to apply the defined data quality management procedures during data collection and preparation.

To explain further, the first step of the framework is the identification of key quality characteristics, followed by the definition of the associated quality requirements for the global CI system and AI/ML subsystem/s that are affected by the training dataset quality. The quality requirements should be based on the category of the system and subsystem, their function, and criticality, as described in ISO 25030 (ISO/IEC 25030, 2019). These requirements are defined for the system quality in use (QIURs) or directly for the system/product characteristics ((PQRs). QIURs and PQRs requirements and corresponding quality measures can be defined based on the quality model in ISO 25010, ISO 25022 and ISO 25023, respectively. In a second phase, the data quality requirements (DQRs) should be specified directly for the data used by the system/subsystem based on the previously identified system requirements, non-functional requirements and constraints, such as data-domain-specific provisions. In addition to that, DQRs can be defined by the data quality Model in the ISO standard 25012 and quality measures in ISO 25024. The differences of purpose between ML/AI data (target data) and other context data (nontarget data) should be taken into account, as well as if their quality impact on the system behavior can be controlled or predicted. This should be considered when setting the strictness of the quality requirements. Afterward, it is important to evaluate the defined quality requirements in order to satisfy them in third step. Prior to data collection and dataset compilation, risks for data quality need to be determined, including the risk associated with human factors, and a method should be specified to mitigate their impact on the data and the system. The three steps above fall within the Requirement Specification and Data Quality Planning stage (first three steps in Figure 2), that outputs a data quality strategy and how to implement it, taking into account the specified data requirements, relevant policies, or domain-related standards. Active management of data quality typically begins only after data collection process (Gong et al., 2023). Once the data quality plan is prepared, the quality implementation procedures for system and human data should be identified, including dataset definition, data acquisition, and quality review as discussed in ISO 22989 as part of the second phase.

In data preparation and data quality monitoring and control, the fourth step in Figure 2 is about addressing and implementing data quality management procedures, following the recommendations associated with the data quality risks identified in the previous step, for each data management stage (a) dataset definition, (b) data acquisition, (c) data annotation, (d) data preparation and cleaning, (e) data augmentation, data sampling, (f) dataset composition, and (g) data quality checking (ISO/IEC 23053, 2022c). The nontarget contextual data may not require such management stages, however at minimum, a data quality check should be performed and considered in improvement efforts of the system quality. The fifth step, is about the identification and report of unresolved data non-conformity and quality issues. Potential impacts on the CI system performance should be analysed and quantified when possible and the data quality plan should also be evaluated for further improvement of the processes. This is the final step of the feedback process to systematically improve the data quality and eliminate identified root causes of data non-conformity. This step



therefore depends on how data is collected to create a datasets and should guide the final dataset collection.

4 Case studies

In this section we demonstrate the application and efficacy of the proposed hybrid standardization approach in managing data quality within two specific case studies in industrial HRI. The case studies were designed and conducted at Irish Manufacturing Research, Ireland. Given that the datasets have not yet been published, we have provided a high level description of the data. We focus on the data quality challenges and management methods used, while offering some practical strategies to ensure good quality CI datasets.

4.1 Human-robot teaching interaction through kinesthetic programming by demonstration (PbD)

This case study investigates human and robot teaching interaction using Programming by Demonstration (PbD). PbD is a supervised method which allows humans to impart skill to a robot without explicitly programming the robot through different direct or indirect input methods. One of the direct input methods is kinesthetic teaching, where operators manually guide the robot's movement to demonstrate specific tasks. The method is suitable for where traditional programming struggles to adapt complex and frequent customization of tasks in HRI. This approach reduces programming setup time and increases the robot's adaptability to varied task requirements (Villani et al., 2018).

In developing our case study, we built upon the experimental framework that was detailed in our previous work Mehak et al., 2023). The framework has been specifically designed for collecting multimodal collaborative data. The purpose of collecting this data is three fold; first, to assess human performance and robot's learning curve in real time as it requires new skills from human demonstration; second, to evaluate the ergonomics and cognitive impact on human operators; and third, to train an ML model to develop an adaptive HRI system. To collect the collaborative data, we used state-of-the-art tools (Lin et al., 2023) and metrics (Skaramagkas et al., 2023) for evaluating human performance and analyzing the robot's task execution algorithms.

The experimental setup of this study involves a collaborative robotic arm with gravity compensation control to accurately mimic human movements, an RGB-D vision sensor for object tracking, and a mediaPipe-based vision system (Ma et al., 2022) is developed to capture human ergonomic data of teaching postures. We use a wearable eye tracker to capture the cognitive load and focus of human operators during their teaching interactions. During the experiment, operators perform two of predefined tasks, demonstrating each step to the robot. Task one is simple object manipulation and the other task involves object sliding over the



slider representing traditional activities on a production line that could benefit from automation. At the end of the experiment, participants are asked to fill out questionnaires to give their subjective feedback. The collected data consists of three sets: human motion data, robot demonstration data and human cognitive data. Further, we applied our hybrid standardization approach and guidelines from data acquisition to dataset compilation, following the procedure illustrated in Figure 2. A comprehensive breakdown of data sources, tasks, and challenges related to data quality that cover the need for high precision and consistency is presented in Table 2.

The following sections detail the implementation of hybrid standardization in our case study.

4.1.1 Quality requirement identification for the system

In this case study, the system consists of two main subsystems; the human subsystem, which is responsible for demonstrating tasks to the robot, and the robot, which learns them and later executes them. The additional subsystems include a human motion tracking system and an eye tracker head system, as previously mentioned. The quality requirements for the motion tracking system are set to capture images at a resolution of 1080p with a frame rate of 60 frames per second, ensuring real-time data processing to accurately track human movements. The eye tracker system is required to operate with a sampling rate of at least 100 Hz and a latency of less than 10 milliseconds to precisely monitor cognitive load and attention. These specific requirements are tailored to the needs of our collaborative robotic training environment, demonstrating that the exact specifications can vary based on the application and system goals. The requirements are identified on the basis of performance efficiency described in (ISO/IEC 25010, 2023a). At the system level, quality requirements focus on ensuring complete data flow and interaction between these subsystems (ISO/IEC 25012, 2008a), scalability to adapt to varying user interactions, and robustness to operate effectively under various situations, as outlined by ISO/IEC 25030 (ISO/IEC 25030, 2019). Some additional requirements are not explicitly covered by these standards, such as real-time adaptation, multimodal data integration, and the scalability necessary for dynamic environmental responsiveness, which are some of the characteristics of collaborative systems (Odong et al., 2022).

4.1.2 Data quality requirements

The quality of data from both human and robotic sources is a key factor in building a reliable HRI system (Mia and Shuford, 2024). This kind of data is often collected in an experimental setting from a diverse group of participants, including some who have never interacted with robots before. The collaborative data for this use case consists of multiple categories of data, including human motion data, robot programming data, user feedback and cognitive aspects of operator recorded during the teaching interaction. The examples of data instances can be visualized in Figure 3. Ensuring the integrity of the data is crucial, as it will be used for system performance assessment and the training of a deep learning model in the future. Therefore, we adhere to ISO/IEC 25012 (ISO/IEC 25012, 2008a) and ISO 8000-61 (ISO 8000-61, 2022a) to define the quality requirements for the data. Data accuracy is one of the important quality requirements for collaborative data (ISO 8000-61, 2022a). In this use case, to ensure accuracy, it is needed to record the actual movement of humans and capture precise positional coordinates, orientation and joint states of the robot. Another quality requirement is defined to ensure data completeness (ISO/IEC 25012, 2008a; ISO 8000-61, 2022a)that no data points are missed and capture a full representation of human and robot interactions. Timeliness is an additional data characteristic (ISO 8000-61, 2022a), with data processed within a 10 m window to provide immediate insights into the operator's cognitive aspects and integrity throughout the data acquisition phase. These requirements are derived based on the functionalities of each subsystem to maintain data quality.

4.1.3 Risk identification and data quality plan

To identify possible risks related to data acquisition of each type of data we referred to the guidelines presented by data quality (ISO 8000-61, 2022a). The primary risks include accuracy risk, where inaccuracies in task demonstration data could lead to misinterpretation of human actions by the robot resulting in incorrect task execution. Moreover, the completeness risks arise from the vision system's occlusion problem. This identification highlight that even a minor deviations in recording human postures can lead to substantial errors such as missing data, which in turns can generate inaccuracies in the analysis, thus affecting the overall validity and reliability of the research findings. The risk related to cognitive data involves precisely measuring and timely processing of cognitive assessments such as engagement levels and cognitive loads, which are critical for enabling real-time adaptive teaching responses. Additionally, the subjective nature of user feedback data requires

Data sources	System operations	Description	Unique challenges in CI	Data quality issues	Impact on system performance	Applicable standards
Human Motion Data	Data collection from vision sensor	Recording of human postures while performing teaching tasks	 Susceptibility to noise, Variability across individualise 	- Issues in data reliability and accuracy	 Impaired evaluation of user performance, Misalignment leading to inefficient task execution and safety issues 	- ISO/IEC 25012 and ISO 800-61
Human cognitive state data	Extracting eye tracking data	Eye-tracking metrics, involving cognitive process and visual attentions recorded during HRI.	 Extensive preprocessing required, Inconsistent performance due to changes in environmental conditions 	Timeliness, and consistence	 Inconsistent performance due to changes in environmental conditions Delays in data processing can impede the real-time interaction 	- ISO/IEC 25012 Data quality model, - ISO 8000-61 Data quality management
Robot data	Data collection from robot sensors	Robot - learned behaviors, which are policies acquired through human demonstrations	- Task adaptation	Accuracy and timeliness	- Reduced system learning efficiency and adaptation efficacy	- ISO/IEC 25012 and ISO/IEC 23053
Unified multimodal data	- Data integration and synchronization	Combined data from different data channels	 Ensuring data coherence across different modalities and managing diverse data sources effectively Disparities in data types can lead to integration issues 	Completeness, integration, and synchronicity	- Incomplete or erroneous responses from the HRI system	_

TABLE 2 Use case 1 summarized: data sources, data quality challenges, impact and applicable standards.

comprehensive design and implementation of feedback methods as they reflect user experiences, thereby supporting accurate system adjustments based on real user feedback.

To manage the defined potential risks in this case study, the data quality measures specified in ISO/IEC 25012 are followed as quality assessment plan.

4.1.4 Quality control in data acquisition

The data acquisition is conducted in a laboratory setting. The participants perform two simple assembly tasks (a) target object placement and (b) object sliding on the slider. While completing the tasks, a head mounted eye tracker system collects participant gaze and pupil dilation. At the same time, the motion tracking system captures the data about their teaching postures which include body's joint movements. In addition, subjective feedback on workload and usability is also recorded using online questionnaires. These responses are context-driven for developing the understanding of HRI interaction dynamics and refining the robot's learning algorithms. After data acquisition, an initial investigation of the data was performed to ensure that the data values and formats were accurate, as a data preparation step in (ISO/IEC 23053, 2022c), and the data are categorised into two groups as user data and robot data.

4.1.5 Dataset preparation and quality control management

After data collection, the first step is to extract data from system specific format to user accessible format for data preprocessing involving data cleaning, where inaccuracies and inconsistencies were removed. This involves cleaning out noise from the eye tracking data and motion data, realigning the posture data, and handling missing or extreme values. Furthermore, cleaning of eye tracking data involves eliminating data points with unrealistic values above an acceptable physiological limit for pupil size and eye movement direction. Non-informative spikes such as blinks and data loss due to poor calibration are retained and filled via linear interpolation. Simultaneously, for motion data, the missing values due to occlusions are addressed using smoothing algorithms to impute the data. The data were annotated to allow for contextual meaning that is crucially needed for the



Visualization of data examples from case study 1. (A) Display of randomly selected data points derived from signals gathered and processed during the Programming by Demonstration (PbD) task, featuring varying sampling rates and temporal dynamics. (B) Heatmap representing the cumulative duration of gaze fixations, superimposed on the robotic setup from a side view. (C) Mapping of human posture trajectories, illustrating movement patterns during the interaction.

preparation of CI data. For instance, annotations pertaining to user data are labeling human postures as "ergonomic" or nonergonomic and cognitive data that have high task engagement levels inferred from the eye tracking data and questionnaire responses. The next step is to combine both data groups into a complete dataset in order to correlate and analyze data from different sources; this step is called data integration in (ISO/IEC 25012, 2008a). Lastly, the data set is transformed into suitable formats to align with the technical requirements of analysis tools and learning models.

After data set compilation, an internal report was produced to highlight key processes and expert feedback on data quality and unresolved issues. Each data streams present specific issue that were reported with action taken from (ISO 8000-61, 2022a).

4.2 Human-in-the-loop telerobotics

The human-in-the-loop teleoperation case-study aims to address flexible and dexterous manufacturing tasks by leveraging the strengths of humans and robots (Figure 4). This type of remote operation of a robot is well suited for medical device manufacturing applications, that may involve the use of dangerous tools, maintaining sterile conditions, and manipulation of miniature and malleable device parts. Moreover, it frequently involves highcustomization and low-volume production constraints. The human operator bears the important role of adaptability and dexterity, reducing the implementation costs compared to a fully automated system.However, this role can be hindered by the teleoperation interface design when it reduces the operator's awareness, presence, or engagement in the task. For optimal performance, a humanin-the-loop teleoperation system should be able to monitor and recognize in real-time the operators' internal state, either cognitive or affective, and adapt accordingly to improve their performance. The system can resort to different adaptation strategies depending on the user needs, such as adjustment of the interface design, changes of the interaction modality or automation level (Dehais et al., 2020). The internal state information can be estimated from different types of human-related data, including behavioural, physical and physiological indicators (Lim et al., 2018). In our CI system, an operator's mental workload recognition function is implemented by a multimodal end-to-end deep learning model, that fuses two physiological data streams (Electroencephalography (EEG) and Eye-tracking).

An experimental teleoperation setup was developed for data collection, including a remote robot station with a six-degree-offreedom robotic arm, a hot-wire mounted on the end-effector, and several cameras to provide different views of the remote environment. On the operator station the graphical user interface is presented on a monitor, providing the operator with the realtime camera streams and other robot state information. For training and implementation of the deep-learning model, the physiological data were collected from the subjects and prepared into a dataset. Moreover, data related to the teleoperation interface, robot kinematics, and task performance are recorded simultaneously to provide context, generate labels, or be used as additional input streams to the adaptation mechanism. Next, we specify the key data quality management processes used to achieve the final goal of dataset creation (including dataset definition, data acquisition, data processing and cleaning), particularly those more relevant for



this type of CI data and application, guided by the proposed hybrid standardization framework.

4.2.1 System quality requirements

The overarching goal of such a CI system, with collaboration between a human operator and a teleoperation robotic system, is to maintain a clean room environment and enhance productivity in high-customization and low-volume production scenarios. These goals require the system to support robot manipulation dexterity and flexibility, while providing higher handling precision and velocity. The implementation of the CI system includes a computer subsystem, a human collaborator/operator, a robotic sub-system, the sensors sub-systems, the interfaces between them, and the relevant environment for the system. The computer component implements an AI sub-system with the goal of monitoring and real-time recognition of the operators' internal cognitive state. The operator state recognition function is implemented by a multimodal model that receives as input EEG and Eye-tracking sensor data, and outputs the level of mental workload. The output of the recognition model is communicated to the teleoperation system, which adapts according to the recognized operator state (following a rule-based approach), as depicted in Figure 4. This AI sub-system is not considered safety critical for medical device manufacturing applications, but it is essential to maintain the productivity of the cell.

The first step of the data quality management guide (Figure 2) involves the identification of quality requirements for the system and sub-systems according to their function, criticality, and the context of use. The most relevant quality characteristics of the CI system to meet the stakeholders' needs in this case study (based on the quality model in (ISO/IEC 25010, 2023a), are performance efficiency, interaction capability, flexibility and reliability. A CI production cell throughput requirement can be set no lower than the manual production throughput, and a performance efficiency measure should be selected to indicate the degree to which the system meets this requirement. As a human-in-the-loop system, the interaction capability measures the degree to which the system is learnable within a specific amount of time, is easy to operate and control, is engaging to the user, and can prevent human error. Considering the high-customization and low-volume production needs, the reliability of the system is important, mainly because it is operational and accessible when required for use, fault tolerant, and flexible to adapt to production parts and production workload variation.

For the AI sub-system instead, functional suitability and compatibility are essential. The sub-system should cover and provide accurate results for all of the specified manufacturing tasks and intended users and should be compatible with the other sub-systems, particularly the sensor systems.

4.2.2 Data quality requirements

To implement the second step of the data quality management guide, we first identified the type of data required for the ML task, then the data quality needs, and the most important quality characteristics to monitor and examine, as advised in ISO 25030 (ISO/IEC 25030, 2019) and ISO 8000-61 (ISO 8000-61, 2022a). The cognitive state recognition component requires synchronized time series signals from biosensors and corresponding human state labels. For this particular use-case, we highlight the importance of accuracy, completeness, consistency, credibility, traceability, and understandability characteristics of the 15 data quality dimensions of the ISO 25012 Data Quality Model (ISO/IEC 25012, 2008a). The accuracy of the training and testing data shall be considered as to how accurately reproduces with high-fidelity the operational data, also required to meet system functional correctness requirements. Some examples of data accuracy measurements are the proportion of outliers in the training dataset compared to the operational data (ISO/IEC 25024, 2015b), the training and operational data distribution difference, or the differences between the acquisition chain used for the development of the model and the one used in operation (Cappi et al., 2021). The compliance with data completeness requirements, and the related data representativeness dimension (Cappi et al., 2021), can be measured by more traditional measures such as the proportion of samples with not null values for a specific attribute/feature (ISO/IEC 25024, 2015b), or more simply as the number of variables and variable values covered by the data from the defined Operational Design Domain (ODD) (Cappi et al., 2021). As the use-case data require extensive pre-processing and the synchronization of multiple data streams, the data collection and processing protocol shall be well annotated and logged, to support credibility and traceability on this high-dimensional data, that is not easily interpreted and understandable by non-experts. Some examples of data credibility measures are the proportion of feature values validated/certified by an expert or specific process (ISO/IEC 25024, 2015b), or the proportion of samples flagged during data collection has potentially affected by a quality degradation source.

4.2.3 Risk identification and data quality plan

Data quality risks were then identified for each of the data quality requirements, and a plan was formulated to prevent or manage the risks, following the third step of the guide. One of the major risks in the dataset definition stage is the non-rigorous definition of the ODD, which does not take into account the manufacturing task to perform or the targeted operational context and consequently introduces bias or collects non-representative samples. A data quality risk specific to this use case is the loss of synchronization between the biosensors, particularly when acquired with different devices and systems. Another major quality risk in CI data acquisition is data contamination by the human operator/agent. Data collection should be tailored towards the target users, e.g., for teleoperation operators, as the skill level of different operators may impact their physiological signals and reactions to different task conditions. Similarly, when the task requires the operator to move, steps should be taken either during data collection or after data collection, to minimize the risk of data quality degradation due to movement artifacts or occlusions. The likelihood and impact of CI data quality issues are task and application dependent and should be estimated accordingly. For example, in industrial tasks carried

out inside a building, with fixed lighting levels and interaction through a screen display or virtual reality headset, such as in the present case study, the light effects on the pupil signal and pupil occlusions are less likely than for tasks carried out outside, or for eye tracking data collected with portable glasses. For EEG data, instead, it is empirically known that muscle and movement artifacts have a bigger impact on the signal quality than heart artifacts, due to their magnitude and lower availability of artifact removal/correction methods. Moreover, for multimodal systems, the impact of quality degradation in one of the modalities is expected to be lower than in an unimodal system (Poria et al., 2017). Paradoxically, during data preparation the quality of the data can be affected by data transformations and the cleaning process. It is crucial that the data processing procedure is planned and well justified beforehand, and not changed during or after this stage to avoid the introduction of bias.

Subsequently, we present some practical examples of how these risks were minimized or handled during data acquisition and data preparation. These examples serve as actionable solutions to manage data quality risks.

4.2.4 Quality control management during data acquisition

Having identified the data sources, data features and data quality needs of the system, as part of a dataset definition step (Picard et al., 2020), data collection was planned and carried out in an experimental setting. During the performance of a teleoperation task, a screen-based eye-tracker system recorded the subject's gaze and pupil changes with a 60 Hz sampling rate, a mobile EEG system recorded the subject's brain activity changes with a 500 Hz sampling rate, along with simultaneous recording of the robot kinematics, task duration, and performance. After each trial/task, the subject was requested to fill in online questionnaires to assess their perceived cognitive state and to help generate operator state labels for the data. To summarize, three groups of related data were acquired: (a) user state data, (b) user behavior data, and (c) user performance data (see Figure 5 for examples of the data collected).

During data acquisition, data quality monitoring and control procedures were implemented for the identified data quality risks, as advised in ISO 8000-61 (ISO 8000-61, 2022a). The EEG and eye tracking signals were synchronized during collection with the Lab Streaming Layer (LSL) (Kothe et al., 2024), using keyboard events to mark the start and end of the tasks. LSL is a standardized protocol that allows any sensor device that uses it to stream different data types and multiple channels in a real-time, unified, and synchronized way. Using this protocol allowed for easier and higher quality data synchronization; however during data collection, we nevertheless monitored the status of the streams and that the event markers were being recorded (through real-time stream visualization), to catch and correct any synchronization and acquisition errors early. If loss of synchronization still happened during the acquisition, a manual synchronization protocol was implemented post-data collection, albeit having lower precision. This issue was flagged when it happened, and during data preparation, its impact on the data streams was analysed (e.g. lower temporal correlation compared to the other recordings).

For our use case, as the data were collected from non-experts, we monitored the subjects during the task to ensure they did not



show signs of confusion about the instructions, as the data recorded in this situation will not be relevant to the ML task. Moreover, we monitored them to detect any potential source of noise or artifacts in the sensor data, coaching the subject to avoid unnecessary movements or moving away from the workstation. This last step is particularly important for the physiological sensors used, as they are sensitive to several subject-related noise and typically have a physical distance limit up to which they can capture the signals. Nonetheless, the collected data should represent the expected movement and range of motion of the human in operational conditions to comply with data representativeness requirements (Cappi et al., 2021). Similarly, for eye-tracking data, special care should be taken to match the expected lighting conditions during operation with those during data collection, when possible to control this factor. When lighting levels are variable, the data can be cleaned during data preparation. However it may not be possible to identify and remove all lighting change effects on the eye-tracking data. Annotations of potential sources of noise and the corresponding time stamps can serve as a preliminary quality report of the data and can aid in the data cleaning and processing step.

The interpretation of CI data, which can include indicators of latent states and partially observable variables, may vary significantly. Therefore, context information should be used for more effective collaborative interactions (Zachary et al., 2015). The user behavioral and performance data was collected and recorded by ROS (a commonly used Robot Operating System), triggered by the same events markers streamed with LSL. The data were added to the final dataset by either synchronizing with the physiological streams (e.g., real-time robot kinematic streams) or by annotating data segments (e.g., annotating the trials with the associated task performance). This context can be used as additional input streams to the model or to label different ML tasks, supporting the reuse of the final dataset to train other human-robot related tasks.

4.2.5 Quality control management during data preparation

Following data acquisition, an initial exploration of the data was performed to confirm that the values and formats are as expected and reasonable for the task - data preparation step in (ISO/IEC 23053, 2022c). Before data cleaning, error rates were computed with respect to NaNs, missing values, invalid data, or outliers, and subsequently, cleaning, filtering, imputation, or discarding of the data was performed according to the acceptable rates (considering how the data streams are segmented for real-time recognition).For example, for a pupil diameter signal with missing data, small gaps can be interpolated over, while long gaps above 40 ms should be marked invalid to avoid introducing unrealistic values (Kret and Sjak-Shie, 2018). The intra- and inter-subject data distribution should be measured to detect inconsistencies within subjects or the population. It should be noted that, for human data, outliers are common and may correspond to a normal variation in the measured signals. We leveraged the data collection annotations and additional context streams to better understand if outlier values were due to normal variability or caused by non-task related sources (e.g. when the subject was distracted or asked for help during the experiment).

10.3389/frobt.2024.1434351

The physiological data was further preprocessed and cleaned according to community domain standards. For EEG data, different approaches have been used in the literature (Kastrati et al., 2021). The validity and usefulness of different EEG preprocessing approaches for the use of deep learning are still being studied (Kingphai and Moshfeghi, 2021), as there is a lack of methods to compare them. For our CI application, as fast stream preprocessing is required (within a few seconds for real-time implementation), we employed an automated minimal processing protocol that applies data resampling, spectral filtering, automated artifact correction with the Artifact Subspace Reconstruction algorithm (Mullen et al., 2015), and average re-referencing. This protocol aims to be feasible in online operational conditions and avoids the need for manual intervention, which can introduce bias and discrepancies between the training and operational data acquisition chain. In our use case, movement during the task was limited and eye-related artifacts were the most common type of contamination. Nonetheless, for tasks that require the subjects to move, a minimal processing protocol may not be enough to remove/correct these artifacts. For eye tracking data, the transformation of the signals may first be needed to retrieve data related to the pupil or gaze. The cleaning of these signals should target periods with unrealistic values or distortions due to blinks or other interference with eye tracking. Specifically for pupil-related data, correction for lighting levels may be necessary to increase signal quality and keep only task-related changes (remove pupil light response effects in pupil size) (Lu et al., 2015).

All streams were then segmented into trials according to the keyboard markers and subsequently into a few-second epochs. This segmentation process varies depending on the data acquisition protocol, but it is important to add the relevant metadata so that the link to the raw data is not lost. For our use case, information about the task condition ID, subject ID and epoch order within the trial was kept. As data were collected from different subjects, different days/sessions, and different streams, it is important to normalize and scale it for each subject separately [see (Guo et al., 2022) for examples of the use of normalization methods to reduce subject differences and biases], particularly for subject-independent state recognition models, as advised in (ISO/IEC 23053, 2022c).

4.2.6 Data quality report

After dataset compilation, all the processing steps are reported along with the data, as well as data quality issues that were not resolved or even when corrected may affect the model's performance. In our use case, we note the identification of synchronization and event marker stream recording errors that were corrected manually, which may introduce variations in the quality of the data. As the effect of these types of errors is not trivial to assess, caution should be taken at the model evaluation stage, estimating the model's output confidence for the affected data samples (model performance may be over- or under-estimated). Additionally, we recommend the continuous assessment of the suitability of the dataset and production model to the operational data and task. As the interaction technology and the non-linear dynamic nature of human-system interactions evolves in complex CI systems (Karwowski, 2012), so does the risk of concept drift and the quality of the CI solution. An ideal CI system should be able to use real-time dynamic data to retrain, update, and improve its behavior continuously according to the new data.

Table 3 presents a summary of the proposed data quality management processes for the teleoperation case-study.

5 Conclusions and impact for future work

This work was inspired by the empirical challenges that were encountered during data acquisition and management, aimed at developing and testing the application of CI, specifically in the HRI area of the CISC project (Leva and Estrada Lugo, 2023). Some of those challenges can affect the ability of collaborative applications to be fully used in industrial settings, potentially limiting their efficiency.In the HRI context, the difficulty in operationalization of relevant human signals can affect the safety of the human-robot interactions and user experience/ergonomic benefits of HRI, while the quality of human feedback can affect the robot's knowledge acquisition ability. The presence of noise, unaccounted individual variation and missing data may limit the system's interaction ability and the implementation of complex system functionalities, such as workspace or workpiece sharing.

Consequently, the article reported on the need for better data integration and data quality management. These issues were generalized, starting with limitations and challenges encountered within the specific case studies. Therefore, it is worth highlighting that they are informed by the technology deployed, the methods used and the software tools available at the time of the experiment. For example, the experimental data collection methods for the telerobotic case study reported in Section 4.2 have several limitations that can have an impact on the quality of the dataset and trained model, namely due to the collection within a controlled lab setting, performing only one specific teleoperation task during data collection, with a limited number of experimental conditions. Even the proprietary software used for EEG and eye tracking data acquisition did not offer a user-friendly protocol for interfacing with each other and did not facilitate the synchronization of the time stamps with other data (no absolute time stamps were recorded). For the case study regarding human-robot teaching interaction in Section 4.1, the realtime feedback from the robot regarding the performance on the quality of the trajectory demonstrated by the user is only observable after an actual execution phase.

The two studies were covered by research ethics approval. However, in general, issues related to the use and monitoring of physiological data, such as those considered in this work, may need to be explored further in connection with the need to monitor human performance and how they can be deployed to avoid possible stigmatization (O'Brolchain and Leva, 2020). Moreover, the impact and relevance of these aspects can be very meaningful, as discussed in Section 2. Following are some important considerations.

• The need for specific human in the loop provisions in safety critical application, required by the EU AI act (Laux et al., 2024), will come in conjunction with the necessity for those applications to demonstrate better integration of their data components to handle the automation/robotic inputs and outputs, the control system inputs and outputs, and the many possible forms of human inputs and outputs such as action, physiological signals and subjective feedback.

Data sources	System operations	Description	Unique challenges in CI	Data quality issues	Impact on system performance	Applicable standards
Data Sources	System Operations	Description	Unique Challenges in CI	Data Quality Issues	Impact on system Performance	Applicable Standards
Human cognitive state data	Sensor data collection	Recording of sensor data as proxy to understand the internal operator state	 High-fidelity reproduction of operational data, Sample representing well a population, Data with noise and artifact contamination 	Accuracy, Representativeness	Lower generalization ability and recognition performance	ISO/IEC 25012 for Accuracy and Reliability
Human cognitive state data	Sensor data processing	Retrieving meaningful latent information from human sensor data	 Extensive preprocessing required, Introduction of bias in the processing stage 	Credibility, Traceability, and Understandability	Difficulty in tracing and identification of data quality root causes for non-experts on the data type	- ISO/IEC 25012 Data quality model, - ISO 8000-61 Data quality management
Multi-modal data	Data synchronization and integration	Combining heterogeneous data types for analysis and processing	Data coherence across different data types	Data consistency, Integration, Synchronicity	Reduced system learning efficiency and adaptation efficacy	_
Human data, with robot and task context	System real-time adaptation to operator	Adjusting to changes in operator state and performance	Real-time system processing of operator data, with robot and task performance context	Data contextualization in unstructured and complex environment	Reduced system flexibility and efficacy in dynamic conditions	_

TABLE 3 Use case 2 summarized: data sources, data quality challenges, impact and applicable standards.

- There is a need for improved customization in HRI. The customization relies on utilizing human data for each operator, adjusting interface preferences and task allocation. Successful implementation requires precise, dependable, and timely data for development and training, as poor quality can lead to task inaccuracies and communication disruptions.
- Finally, these CI data quality and management issues are to be supported by other data-related organizational processes, such as the management of multimodal data storage architecture, data transfer, data operations, and security. In addition, organizational structure and human resources are key to implementing and maintaining data quality policies and cultures. These can be subject to other industry standards and guidelines not considered in the present study (which was limited to ISO standards), such as those issued by the International Society of Automation.

5.1 Summary of key findings

The key recommendations derived in the present paper can be summarized as follows.

• The use of standardized protocols can ensure seamless integration and processing of multimodal data and address challenges related to data format discrepancies.

- Maintaining robust metadata management practices and documenting the origin, context, and preprocessing history of data can facilitate its traceability and reliability in diverse analytical scenarios.
- Understanding the data types, heterogeneous data sources, nature of data (qualitative or quantitative), and intended use of data within the CI context is of the utmost importance. This understanding provides transparency in system behavior which is useful in ensuring data quality in accordance with established data quality standards.
- CI systems are not immutable and are characterized by their dynamic and adaptive nature, as discussed in both the use cases presented in this paper and the latest literature (Schleiger et al., 2024). We must foster a culture of continuous improvement, where feedback is actively sought and used to refine data processes, improve system performance, and address any emerging issues (such as possible changes in paces and interaction as the level of experience and familiarity between the system and the operator increases over time.

Recent works have started to address data-related issues and their impact on the performance and quality of ML/AI data, mostly focused on mitigating the risk of small, noisy, biased or corrupted datasets through data resampling methods or model optimization for increased robustness to specific quality issues (Whang et al., 2023). The standardization of the data quality management process can help in the systematic prevention of quality degradation, thereby contributing to the improvement of the safety and behavioural performance of data-driven and collaborative systems. Furthermore, with the Artificial Intelligence Act (Laux et al., 2024) coming into force, proposed by the European Commission to regulate AI in the EU, this framework can be used to put a data quality management system in place that ensures compliance with this regulation and supports traceability, accountability and fairness requirements. Finally, taking into consideration the narrow definition of data quality employed in the proposed hybrid standardization framework, that highly depends on the purpose and conditions of the data use, its application to foundation and general purpose AI models is limited. Further work is needed to understand how the lack of a clear link between data purpose and data quality affects these models' quality and trustworthiness, the ethical collection and use of personal data (principle of data minimisation), and the responsible use of data storage resources.

Future research will aim to explore other standards relevant to collaborative systems in other domains, such as automation for the process industry and energy sector related to control rooms and distributed control systems. For prospective collaborative intelligence and decision-support applications, paying attention to aspects similar to those discussed in this paper may also be required.

Data availability statement

The original contributions presented in the study are included in the article; further inquiries can be directed to the corresponding authors.

Author contributions

SM: Conceptualization, Data curation, Formal Analysis, Investigation, Methodology, Project administration, Resources, Software, Validation, Writing–original draft, Writing–review and editing. IR: Data curation, Formal Analysis, Investigation, Methodology, Software, Visualization, Writing–original draft, Writing–review and editing. KS: Software, Writing–review and editing. AR: Writing–review and editing. JK: Supervision, Writing–review and editing. MG: Project administration, Resources, Supervision, Validation, Writing–review and editing. GG: Funding acquisition, Supervision, Writing–review and editing. ED: Supervision, Writing–review and editing, Funding acquisition. ML: Writing–review and editing, Project administration, Resources, Supervision, Validation, Funding acquisition.

References

Funding

The author(s) declare that financial support was received for the research, authorship, and/or publication of this article. This work was fully funded by the Horizon 2020 Research and Innovation Program of the European Union under the Marie Skłodowska-Curie grant agreement no. 955901 https://www.ciscproject.eu/. The work was partially supported by the MUSA-Multilayered Urban Sustainability Action project, funded by the European Union-NextGenerationEU, under the Mission 4 Component 2 Investment Line of the National Recovery and Resilience Plan (NRRP) Mission 4 Component 2 Investment Line 1.5: Strengthening of research structures and creation of R&D "innovation ecosystems," set up of "territorial leaders in R&D" (CUP G43C22001370007, Code ECS00000037); Program "piano sostegno alla ricerca" PSR and the PSR-GSA-Linea 6; Project ReGAInS (code 2023-NAZ-0207/DIP-ECC-DISCO23), funded by the Italian University and Research Ministry, within the Excellence Departments program 2023-2027 (law 232/2016); and-FAIR-Future Artificial Intelligence Research-Spoke 4-PE00000013-D53C22002380006, funded by the European Union-Next Generation EU within the project NRPP M4C2, Investment 1.,3 DD. 341, 15 March 2022. The work of Kelleher was also supported by the ADAPT Research Centre, which is funded by Science Foundation Ireland (Grant 13/RC/2106 P2) and is cofunded by the European Regional Development Fund. The second case study presented in this article is part of a project that has received funding from the European Union's Horizon 2020 research and innovation programme under the Marie Skłodowska-Curie grant agreement No. 847402.

Conflict of interest

Authors Shakra Mehak and Michael Gulifoyle were employed by Pilz Ireland Industrial Automation.

The remaining authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

An, J., Cheng, J., Gui, X., Zhang, W., Liang, D., Gui, R., et al. (2020). A lightweight blockchain-based model for data quality assessment in crowdsensing. *IEEE Trans. Comput. Soc. Syst.* 7, 84–97. doi:10.1109/tcss.2019.2956481

Angleraud, A., Sefat, A. M., Netzev, M., and Pieters, R. (2021). Coordinating shared tasks in human-robot collaboration by commands. *Front. Robotics AI* 8, 734548. doi:10.3389/FROBT.2021.734548

Admoni, H., and Scassellati, B. (2014) "Data-driven model of nonverbal behavior for socially assistive human-robot interactions," in *Proceedings of the 16th international conference on multimodal interaction*, 196–199.

Amershi, S., Weld, D., Vorvoreanu, M., Fourney, A., Nushi, B., Collisson, P., et al. (2019). "Guidelines for human-ai interaction," in *Proceedings of the 2019 chi conference on human factors in computing systems*, 1–13.

Batini, C., Cappiello, C., Francalanci, C., and Maurino, A. (2009). Methodologies for data quality assessment and improvement. *ACM Comput. Surv.* 41, 1–52. doi:10.1145/1541880.1541883

Bekele, E., and Sarkar, N. (2014). "Psychophysiological feedback for adaptive human-robot interaction (hri)," in *Advances in physiological computing*. Springer, 141-167.

Bissoli, A., Lavino-Junior, D., Sime, M., Encarnação, L., and Bastos-Filho, T. (2019). A human–machine interface based on eye tracking for controlling and monitoring a smart home using the internet of things. *Sensors* 19, 859. doi:10.3390/s19040859

[Dataset] Cappi, C., Chapdelaine, C., Gardes, L., Jenn, E., Lefevre, B., Picard, S., et al. Dataset definition standard (dds) (2021).

Celiktutan, O., Skordos, E., and Gunes, H. (2017). Multimodal human-human-robot interactions (mhhri) dataset for studying personality and engagement. *IEEE Trans. Affect. Comput.* 10, 484–497. doi:10.1109/taffc.2017.2737019

Dehais, F., Lafont, A., Roy, R., and Fairclough, S. (2020). A neuroergonomics approach to mental workload, engagement and human performance. *Front. Neurosci.* 14, 268. doi:10.3389/fnins.2020.00268

Dellermann, D., Calma, A., Lipusch, N., Weber, T., Weigel, S., and Ebel, P. (2021). The future of human-ai collaboration: a taxonomy of design knowledge for hybrid intelligence systems. *arXiv Prepr. arXiv:2105.03354*. doi:10.48550/arXiv.2105.03354

Demir, K. A., Döven, G., and Sezen, B. (2019). Industry 5.0 and human-robot co-working. *Procedia Comput. Sci.* 158, 688–695. doi:10.1016/j.procs.2019.09.104

Eckman, S., Plank, B., and Kreuter, F. (2024). The science of data collection: insights from surveys can improve machine learning models. *arXiv Prepr. arXiv:2403.01208*.

Ehrlinger, L., and Wöß, W. (2022). A survey of data quality measurement and monitoring tools. *Front. big data* 5, 850611. doi:10.3389/fdata.2022.850611

Gao, Q., Liu, J., and Ju, Z. (2021). Hand gesture recognition using multimodal data fusion and multiscale parallel convolutional neural network for human-robot interaction. *Expert Syst.* 38, e12490. doi:10.1111/exsy.12490

Gast, J., Bannat, A., Rehrl, T., Wallhoff, F., Rigoll, G., Wendt, C., et al. (2009). "Realtime framework for multimodal human-robot interaction," in 2009 2nd conference on human system interactions (IEEE), 276–283.

George Kokotinis, Z. A. G. M., Makris, S., and Arkouli, Z. (2023). On the quantification of human-robot collaboration quality. *Int. J. Comput. Integr. Manuf.* 36, 1431–1448. doi:10.1080/0951192X.2023.2189304

Gong, Y., Liu, G., Xue, Y., Li, R., and Meng, L. (2023). A survey on dataset quality in machine learning. *Inf. Softw. Technol.* 162, 107268. doi:10.1016/j.infsof.2023.107268

Gudivada, V., Apon, A., and Ding, J. (2017). Data quality considerations for big data and machine learning: Going beyond data cleaning and transformations. *Int. J. Adv. Softw.* 10, 1–20.

Guo, Y., Freer, D., Deligianni, F., and Yang, G. Z. (2022). Eye-tracking for performance evaluation and workload estimation in space telerobotic training. *IEEE Trans. Human-Machine Syst.* 52, 1–11. doi:10.1109/THMS.2021.3107519

Gupta, N., Patel, H., Afzal, S., Panwar, N., Mittal, R. S., Guttula, S., et al. (2021). Data quality toolkit: Automatic assessment of data quality and remediation for machine learning datasets. *arXiv Prepr. arXiv:2108.05935*.

Han, Z., and Williams, T. Towards formalizing hri data collection processes (2022).

Heinisch, J. S., Kirchhoff, J., Busch, P., Wendt, J., von Stryk, O., and David, K. (2024). Physiological data for affective computing in hri with anthropomorphic service robots: the affect-hri data set. *Sci. Data* 11, 333. doi:10.1038/s41597-024-03128-z

Hentout, A., Aouache, M., Maoudj, A., and Akli, I. (2019). Human–robot interaction in industrial collaborative robotics: a literature review of the decade 2008–2017. *Adv. Robot.* 33, 764–799. doi:10.1080/01691864.2019.1636714

Hoffman, G., and Zhao, X. (2020). A primer for conducting experiments in humanrobot interaction. *ACM Trans. Hum.-Robot Interact.* 10, 31. doi:10.1145/3412374

ISO 8000-1 (2022b). 2022 - data quality - Part 1: overview, principles, and terminology. Available at: https://www.iso.org/standard/81745.html.

ISO 8000-61 (2022a). Data quality – Part 61: data quality management: process reference model. Available at: https://www.iso.org/standard/81745.html.

ISO 8000-8 (2015a). 2015 - data quality – Part 8: information and data quality: concepts and measuring. Available at: https://www.iso.org/standard/60805.html.

ISO/IEC 22989 (2023b). Information technology – artificial intelligence – artificial intelligence concepts and terminology. Available at: https://www.iso. org/standard/72116.html.

ISO/IEC 23053 (2022c). Framework for artificial intelligence (AI) systems using machine learning (ML). Available at: https://www.iso.org/standard/74438.html.

ISO/IEC 25000 (2014a). 2014 - systems and software engineering – systems and software quality requirements and evaluation (SQuaRE) – guide to SQuaRE. Available at: https://www.iso.org/standard/35733.html.

ISO/IEC 25001 (2014b). 2014 - systems and software engineering – systems and software quality requirements and evaluation (SQuaRE) – planning and management. Available at: https://www.iso.org/standard/35734.html.

ISO/IEC 25010 (2023a). 2023 - systems and software engineering – systems and software quality requirements and evaluation (SQuaRE) – product quality model. Available at: https://www.iso.org/standard/78176.html.

ISO/IEC 25012 (2008a). Data quality model. Available at: https://www.iso. org/standard/35736.html.

ISO/IEC 25012 (2008b). Systems and software engineering. Available at: https://www.iso.org/standard/35746.html.

ISO/IEC 25024 (2015b). Systems and software engineering. Available at: https:// www.iso.org/standard/35749.html.

ISO/IEC 25030 (2019). Syst. Softw. Eng. – Syst. Softw. Qual. Requir. Eval. (SQuaRE) – Qual. Requir. Framew. Available at: https://www.iso.org/standard/72116.html.

Karat, J., Karat, C. M., and Brodie, C. (2007). "Human-computer interaction viewed from the intersection of privacy, security, and trust," in *The human-computer interaction Handbook*. Boca Raton, FL, United States: CRC Press, 665–684.

Karwowski, W. (2012). A review of human factors challenges of complex adaptive systems: Discovering and understanding chaos in human performance. *Hum. Factors* 54, 983–995. doi:10.1177/0018720812467459

Kastrati, A., Plomecka, M. B., Pascual, D., Wolf, L., Gillioz, V., Wattenhofer, R., et al. (2021) "EEGEyenet: a simultaneous electroencephalography and eye-tracking dataset and benchmark for eye movement prediction," in *Thirty-fifth conference on Neural information processing systems datasets and Benchmarks track (Round 1).*

Khamassi, M., Velentzas, G., Tsitsimis, T., and Tzafestas, C. (2018). Robot fast adaptation to changes in human engagement during simulated dynamic social interaction with active exploration in parameterized reinforcement learning. *IEEE Trans. Cognitive Dev. Syst.* 10, 881–893. doi:10.1109/tcds.2018.2843122

Kim, S. K., Kirchner, E. A., Schloßmüller, L., and Kirchner, F. (2020). Errors in human-robot interactions and their effects on robot learning. *Front. Robotics AI* 7, 558531. doi:10.3389/frobt.2020.558531

Kingphai, K., and Moshfeghi, Y. (2021). On EEG preprocessing role in deep learning effectiveness for mental workload classification. *Commun. Comput. Inf. Sci.* 1493 CCIS, 81–98. doi:10.1007/978-3-030-91408-0_6

Kothe, C., Shirazi, S. Y., Stenner, T., Medine, D., Boulay, C., Grivich, M. I., et al. (2024). The lab streaming layer for synchronized multimodal recording. *bioRxiv*. doi:10.1101/2024.02.13.580071

Kret, M., and Sjak-Shie, E. (2018). Preprocessing pupil size data: guidelines and code, Behav. Res. Methods, 51 1336–1342. doi:10.3758/s13428-018-1075-y

Laux, J., Wachter, S., and Mittelstadt, B. (2024). Trustworthy artificial intelligence and the European Union ai act: on the conflation of trustworthiness and acceptability of risk. *Regul. and Gov.* 18, 3–32. doi:10.1111/rego.12512

Leva, M. C., and Estrada Lugo, H. D. (2023). "Collaborative intelligence and human-ai teaming in safety critical applications: key challenges and opportunities," in *Proceedings of the 33rd European safety and reliability Conference*. Editors M. P. Brito, T. Aven, P. Baraldi, M. Čepin, and E. Zio (Singapore: Research Publishing). doi:10.3850/981-973-0000-00-0.ESREL2023Paper

Li, T. W., Jo, Y., Brumbaugh, Z., Lee, H., Murray, M., and Björling, E. A. (2023) "Low-effort, unqualified, and malicious user behaviors in online human-robot interaction studies," in *Proceedings of the 41st ACM international conference on design* of communication, 63–69.

Lim, Y., Gardi, A., Sabatini, R., Ramasamy, S., Kistan, T., Ezer, N., et al. (2018). Avionics human-machine interfaces and interactions for manned and unmanned aircraft. *Prog. Aerosp. Sci.* 102, 1–46. doi:10.1016/j.paerosci.2018.05.002

Lin, Y., Jiao, X., and Zhao, L. (2023). Detection of 3d human posture based on improved mediapipe. J. Comput. Commun. 11, 102–121. doi:10.4236/jcc.2023.112008

Liu, P., Glas, D. F., Kanda, T., and Ishiguro, H. (2016). Data-driven hri: learning social behaviors by example from human-human interaction. *IEEE Trans. Robotics* 32, 988–1008. doi:10.1109/tro.2016.2588880

Lu, Y., Zheng, W. L., Li, B., and Lu, B. L. (2015) *Combining eye movements and eeg to enhance emotion recognition (AAAI Press)*, 15. Washington, DC, United States: IJCAI', 1170–1176.

Ma, J., Ma, L., Ruan, W., Chen, H., and Feng, J. (2022) "A wushu posture recognition system based on mediapipe," in 2022 2nd international conference on information technology and Contemporary Sports (TCS). IEEE, 10–13.

McGrath, M. J., Duenser, A., Lacey, J., and Paris, C. (2024). Collaborative humanai trust (chai-t): a process framework for active management of trust in human-ai collaboration. arXiv Prepr. arXiv:2404.01615.

Mehak, S., Jain, A., Kelleher, J. D., Long, P., Guilfoyle, M., and Leva, M. C. (2023). "Understanding and quantifying human factors in programming from demonstration: a user study proposal," in *Proceedings of the 33rd European safety and reliability conference (ESREL 2023) (Southampton, UK: European safety and reliability association).* doi:10.3850/978-981-18-8071-1_P241-cd

Mehta, S. A., and Losey, D. P. (2023). Unified learning from demonstrations, corrections, and preferences during physical human-robot interaction. *ACM Trans. Human-Robot Interact.* 13, 1–25. doi:10.1145/3623384

Mentzas, G., Lepenioti, K., Bousdekis, A., and Apostolou, D. (2021). "Data-driven collaborative human-ai decision making. Responsible AI and analytics for an ethical and inclusive digitized society," in 20th IFIP WG 6.11 conference on e-business, e-services and e-society, I3Z 2021, galway, Ireland, september 1–3, 2021, proceedings 20. Springer, 120–131.

Mia, M. R., and Shuford, J. (2024). Exploring the synergy of artificial intelligence and robotics in industry 4.0 applications. J. Artif. Intell. General Sci. (JAIGS) ISSN 1, 3006–4023 1. doi:10.60087/jaigs.v1i1.31

Mosqueira-Rey, E., Hernández-Pereira, E., Alonso-Ríos, D., Bobes-Bascarán, J., and Fernández-Leal, Á. (2023). Human-in-the-loop machine learning: a state of the art. *Artif. Intell. Rev.* 56, 3005–3054. doi:10.1007/s10462-022-10246-w

Moustakis, V. S., and Herrmann, J. (1997). Where do machine learning and human-computer interaction meet? *Appl. Artif. Intell.* 11, 595–609. doi:10.1080/088395197117948

Mullen, T. R., Kothe, C. A., Chi, Y. M., Ojeda, A., Kerth, T., Makeig, S., et al. (2015). Real-time neuroimaging and cognitive monitoring using wearable dry eeg. *IEEE Trans. Biomed. Eng.* 62, 2553–2567. doi:10.1109/tbme.2015.2481482

Nardo, M., Forino, D., and Murino, T. (2020). The evolution of man-machine interaction: the role of human in industry 4.0 paradigm. *Prod. and Manuf. Res.* 8, 20–34. doi:10.1080/21693277.2020.1737592

O'Brolchain, F., and Leva, M. C. (2020). "Getting personal: physiological measurements and human performance in safety critical tasks. facing the ethical implications," in *Proceedings of the 30th European safety and reliability conference and the 15th probabilistic safety assessment and management conference*. Editors P. Baraldi, F. Di Maio, and E. Zio (Singapore: Research Publishing). doi:10.3850/978-981-14-8593-0

Odong, L. A., Perini, A., and Susi, A. (2022). "Requirements engineering for collaborative artificial intelligence systems: a literature survey," in *International conference on research challenges in information science*. Springer, 409–425.

Panagou, S., Neumann, W. P., and Fruggiero, F. (2024). A scoping review of human robot interaction research towards industry 5.0 human-centric workplaces. *Int. J. Prod. Res.* 62, 974–990. doi:10.1080/00207543.2023.2172473

Panduwiyasa, H., Saputra, M., Azzahra, Z., and Aniko, A. (2021). Accounting and smart system: functional evaluation of iso/iec 25010: 2011 quality model (a case study). *IOP Conf. Ser. Mater. Sci. Eng.* 1092, 012065. doi:10.1088/1757-899x/ 1092/1/012065

Perez-Castillo, R., Carretero, A. G., Caballero, I., Rodriguez, M., Piattini, M., Mate, A., et al. (2018). Daqua-mass: an iso 8000-61 based data quality management methodology for sensor data. *Sensors* 18, 3105. doi:10.3390/s18093105

Picard, S., Chapdelaine, C., Cappi, C., Gardes, L., Jenn, E., Lefevre, B., et al. (2020). "Ensuring dataset quality for machine learning certification," in *Proceedings - 2020 IEEE 31st international symposium on software reliability engineering workshops, ISSREW* 2020, 275–282. doi:10.1109/ISSREW51248.2020.00085

Poria, S., Cambria, E., Bajpai, R., and Hussain, A. (2017). A review of affective computing: from unimodal analysis to multimodal fusion. *Inf. Fusion* 37, 98–125. doi:10.1016/j.inffus.2017.02.003

Rangineni, S. (2023). An analysis of data quality requirements for machine learning development pipelines frameworks. *Int. J. Comput. Trends Technol.* 71, 16–27. doi:10.14445/22312803/ijctt-v71i8p103

Ravishankar, J., Doering, M., and Kanda, T. (2024). Zero-shot learning to enable error awareness in data-driven hri. *Proc. 2024 ACM/IEEE Int. Conf. Human-Robot Interact.*, 592–601. doi:10.1145/3610977.3634940

Rožanec, J. M., Novalija, I., Zajec, P., Kenda, K., Tavakoli Ghinani, H., Suh, S., et al. (2023). Human-centric artificial intelligence architecture for industry 5.0 applications. *Int. J. Prod. Res.* 61, 6847–6872. doi:10.1080/00207543.2022.2138611

Saheb Jam, G., Rhim, J., and Lim, A. (2021). Developing a data-driven categorical taxonomy of emotional expressions in real world human robot

interactions. Companion 2021 ACM/IEEE Int. Conf. human-robot Interact., 479-483. doi:10.1145/3434074.3447218

Schleiger, E., Mason, C., Naughtin, C., Reeson, A., and Paris, C. (2024). Collaborative intelligence: a scoping review of current applications. *Appl. Artif. Intell.* 38, 2327890. doi:10.1080/08839514.2024.2327890

Semeraro, F., Griffiths, A., and Cangelosi, A. (2023). Human–robot collaboration and machine learning: a systematic review of recent research. *Robotics Computer-Integrated Manuf.* 79, 102432. doi:10.1016/j.rcim.2022.102432

Sharifi, M., Azimi, V., Mushahwar, V. K., and Tavakoli, M. (2021). Impedance learning-based adaptive control for human-robot interaction. *IEEE Trans. Control Syst. Technol.* 30, 1345–1358. doi:10.1109/tcst.2021.3107483

Skantze, G., Hjalmarsson, A., and Oertel, C. (2014). Turn-taking, feedback and joint attention in situated human-robot interaction. *Speech Commun.* 65, 50–66. doi:10.1016/j.specom.2014.05.005

Skaramagkas, V., Giannakakis, G., Ktistakis, E., Manousos, D., Karatzanis, I., Tachos, N., et al. (2023). Review of eye tracking metrics involved in emotional and cognitive processes. *IEEE Rev. Biomed. Eng.* 16, 260–277. doi:10.1109/RBME.2021.3066072

Teh, H. Y., Kempa-Liehr, A. W., and Wang, K. I. K. (2020). Sensor data quality: a systematic review. J. Big Data 7, 11. doi:10.1186/s40537-020-0285-1

Toniolo, A., Cerutti, F., Norman, T. J., Oren, N., Allen, J. A., Srivastava, M., et al. (2023). Human-machine collaboration in intelligence analysis: an expert evaluation. *Intelligent Syst. Appl.* 17, 200151. doi:10.1016/j.iswa.2022.200151

Trakadas, P., Simoens, P., Gkonis, P., Sarakis, L., Angelopoulos, A., Ramallo-González, A. P., et al. (2020). An artificial intelligence-based collaboration approach in industrial iot manufacturing: key concepts, architectural extensions and potential applications. *Sensors* 20, 5480. doi:10.3390/s20195480

Van Wegberg, M. (2004). Standardization process of systems technologies: creating a balance between competition and cooperation. *Technol. Analysis and Strategic Manag.* 16, 457–478. doi:10.1080/0953732042000295784

Victorelli, E. Z., Dos Reis, J. C., Hornung, H., and Prado, A. B. (2020). Understanding human-data interaction: literature review and recommendations for design. *Int. J. human-computer Stud.* 134, 13–32. doi:10.1016/j.ijhcs.2019.09.004

Villani, V., Pini, F., Leali, F., and Secchi, C. (2018). Survey on human-robot collaboration in industrial settings: safety, intuitive interfaces and applications. *Mechatronics* 55, 248–266. doi:10.1016/j.mechatronics.2018.02.009

Vössing, M., Kühl, N., Lind, M., and Satzger, G. (2022). Designing transparency for effective human-ai collaboration. *Inf. Syst. Front.* 24, 877–895. doi:10.1007/s10796-022-10284-3

Whang, S. E., Roh, Y., Song, H., and Lee, J. G. (2023). Data collection and quality challenges in deep learning: a data-centric AI perspective. *VLDB J.* 32, 791–813. doi:10.1007/s00778-022-00775-9

Yang, Y., Ding, Z., Wang, R., Modares, H., and Wunsch, D. C. (2021). Data-driven human-robot interaction without velocity measurement using off-policy reinforcement learning. *IEEE/CAA J. Automatica Sinica* 9, 47–63. doi:10.1109/jas.2021.1004258

Zachary, W., Johnson, M., Hoffman, R., Thomas, T., Rosoff, A., and Santarelli, T. (2015). A context-based approach to robot-human interaction. *Procedia Manuf.* 3, 1052–1059. doi:10.1016/j.promfg.2015.07.171

Zheng, P., Li, S., Fan, J., Li, C., and Wang, L. (2023). A collaborative intelligencebased approach for handling human-robot collaboration uncertainties. *CIRP Ann.* 72, 1–4. doi:10.1016/j.cirp.2023.04.057

Zhou, G., Luo, J., Xu, S., and Zhang, S. (2022). A cooperative shared control scheme based on intention recognition for flexible assembly manufacturing. *Front. Neurorobotics* 16, 850211. doi:10.3389/FNBOT.2022.850211

Zhou, J., Gandomi, A. H., Chen, F., and Holzinger, A. (2021). Evaluating the quality of machine learning explanations: a survey on methods and metrics. *Electronics* 10, 593. doi:10.3390/electronics10050593